EDITORIALS

7. Lin SM, Lin CJ, Lin CC, Hsu CW, Chen YC. Radiofrequency ablation improves prognosis compared with ethanol injection for hepatocellular carcinoma < or =4 cm. *Gastroenterology*. 2004;127(6):1714-1723.

8. Lu DS, Yu NC, Raman SS, et al. Radiofrequency ablation of hepatocellular carcinoma: treatment success as defined by histologic examination of the explanted liver. *Radiology*. 2005;234(3):954-960.

9. Shiina S, Teratani T, Obi S, et al. A randomized controlled trial of radiofrequency ablation with ethanol injection for small hepatocellular carcinoma. *Gastroenterology*. 2005;129(1):122-130.

Llovet JM, Real MI, Montana X, et al. Arterial embolisation or chemoembolisation versus symptomatic treatment in patients with unresectable hepatocellular carcinoma: a randomised controlled trial. *Lancet*. 2002;359(9319):1734-1739.
Lo CM, Ngan H, Tso WK, et al. Randomized controlled trial of transarterial lipiodol chemoembolization for unresectable hepatocellular carcinoma. *Hepatology*. 2002;35(5):1164-1171.

12. Groupe d'Etude et de Traitement du Carcinome Hepatocellulaire. A comparison of lipiodol chemoembolization and conservative treatment for unresectable hepatocellular carcinoma. *N Engl J Med.* 1995;332(19):1256-1261.

13. Madden MV, Krige JE, Bailey S, et al. Randomised trial of targeted chemotherapy with lipiodol and 5-epidoxorubicin compared with symptomatic treatment for hepatoma. *Gut.* 1993;34(11):1598-1600.

14. Pelletier G, Ducreux M, Gay F, et al. Treatment of unresectable hepatocellular carcinoma with lipiodol chemoembolization: a multicenter randomized trial. *J Hepatol*. 1998;29(1):129-134.

15. Pelletier G, Roche A, Ink O, et al. A randomized trial of hepatic arterial chemoembolization in patients with unresectable hepatocellular carcinoma. *J Hepatol*. 1990;11(2):181-184. Yamasaki T, Kurokawa F, Shirahashi H, Kusano N, Hironaka K, Okita K. Percutaneous radiofrequency ablation therapy for patients with hepatocellular carcinoma during occlusion of hepatic blood flow: comparison with standard percutaneous radiofrequency ablation therapy. *Cancer.* 2002;95(11):2353-2360.
Bloomston M, Binitie O, Fraiji E, et al. Transcatheter arterial chemoemboliza-

17. Bloomston M, Binitie O, Fraiji E, et al. Transcatheter arterial chemoembolization with or without radiofrequency ablation in the management of patients with advanced hepatic malignancy. *Am Surg.* 2002;68(9):827-831.

 Veltri A, Moretto P, Doriguzzi A, Pagano E, Carrara G, Gandini G. Radiofrequency thermal ablation (RFA) after transarterial chemoembolization (TACE) as a combined therapy for unresectable non-early hepatocellular carcinoma (HCC). *Eur Radiol.* 2006;16(3):661-669.

19. Yamakado K, Nakatsuka A, Akeboshi M, Shiraki K, Nakano T, Takeda K. Combination therapy with radiofrequency ablation and transcatheter chemoembolization for the treatment of hepatocellular carcinoma: short-term recurrences and survival. Oncol Rep. 2004;11(1):105-109.

20. Llovet J, Ricci S, Mazzaferro V, et al. Sorafenib improves survival in advanced hepatocellular carcinoma (HCC): results of a phase III randomized placebocontrolled trial (SHARP trial). *J Clin Oncol*. 2007;25(18S):LBA1.

21. Carlomagno F, Anaganti S, Guida T, et al. BAY 43-9006 inhibition of oncogenic RET mutants. J Natl Cancer Inst. 2006;98(5):326-334.

22. Wilhelm SM, Carter C, Tang L, et al. BAY 43-9006 exhibits broad spectrum oral antitumor activity and targets the RAF/MEK/ERK pathway and receptor tyrosine kinases involved in tumor progression and angiogenesis. *Cancer Res.* 2004; 64(19):7099-7109.

23. Hakimé A, Hines-Peralta A, Peddi H, et al. Combination of radiofrequency ablation with antiangiogenic therapy for tumor ablation efficacy: study in mice. *Radiology*. 2007;244(2):464-470.

The Great Debate of 2008—How Low to Go in Preventive Cardiology?

Eric D. Peterson, MD, MPH

Tracy Y. Wang, MD, MS

HE DEBATES OF 2008 HAVE ALREADY BEEN QUITE INtense. During this election year, politicians and pundits alike, reviewing the same set of information, have formulated remarkably different conclusions and recommendations for national policy. The field of preventive cardiology has likewise been witness to its own debate. Spurred by a series of important yet somewhat unexpected clinical trial results, the question of "how low to go" in cardiovascular risk-factor modification has been hotly disputed.

This debate is not new and traditionally has been waged between the "true believers," those with a strong a priori conviction that more aggressive pharmacological treatment will reduce future events, and the "therapeutic nihilists," those who require unequivocal proof before acceptance. In recent years, the true believers have had the upper hand. Epidemiologic data have consistently concluded that lower levels of lipids, blood pressure, and glucose all correlate with less cardiovascular disease. Similarly, among patients with established cardiovascular disease, intensive lipid lowering with statins has been demonstrated to reduce future cardiac events.¹ Thus, national treatment guidelines

See also p 1678.

have progressively lowered their thresholds for initiation of drug therapy as well as the target levels to be achieved.^{2,3} Yet the benefit of aggressive pharmacological therapy for primary prevention is less clear, even among high-risk subgroups.^{4,5} Additionally, while statin therapy appears beneficial for hypertensive patients,⁶ the ideal targets for low-density lipoprotein cholesterol (LDL-C) or blood pressure lowering in these patients have not been defined. Here lies the doubt of the nihilists: "Where is the evidence that intensive lowering is necessarily better or even safe?"

In this issue of *JAMA*, Howard and colleagues⁷ report the results of the Stop Atherosclerosis in Native Diabetics Study (SANDS), which compared aggressive therapy of systolic blood pressure and LDL-C lowering to standard therapy among American Indian patients with type 2 diabetes mellitus. This is one of the first studies to assess the role of aggressive risk factor modification in a high-risk primary prevention setting. The study was well-designed and rigorously conducted with patient follow-up every 3 months for up to 3 years. The authors also examined these questions in a traditionally understudied population. The results showed that patients receiving intensive management had significant regression of carotid intimal medial

©2008 American Medical Association. All rights reserved.

Author Affiliations: Duke Clinical Research Institute, Duke University Medical Center, Durham, North Carolina. Dr Peterson is also Contributing Editor, *JAMA*. Corresponding Author: Eric D. Peterson, MD, MPH, Duke Clinical Research Institute, PO Box 17969, Durham, NC 27715 (peter016@mc.duke.edu).

¹⁷¹⁸ JAMA, April 9, 2008—Vol 299, No. 14 (Reprinted)

thickness (IMT) and reduced left ventricular mass compared with patients treated with standard strategies. In contrast to the Anglo-Scandinavian Cardiac Outcomes Trial (ASCOT),⁸ there appeared to be no synergistic effect between lipid and blood pressure lowering on these separate surrogates-aggressive LDL-C lowering was associated with IMT regression, but was not a predictor of left ventricular mass reduction; conversely, aggressive systolic blood pressure lowering was associated with left ventricular mass reduction, but not with IMT regression. Importantly, while the study was not powered to address clinical outcomes, intensively managed patients had no significant reduction in cardiovascular events after 3 years of follow-up relative to patients receiving standard treatment. In fact, more adverse effects (hypotension, hyperkalemia) were observed in the group receiving aggressive blood pressure treatment.

Some of the obstacles facing primary prevention trials are highlighted in SANDS. Given low annual event rates, primary prevention trials often require large sample sizes and long-term follow-up to assess hard clinical outcomes. Comparing event differences becomes particularly challenging when the question moves from a placebo-based comparison to one that assesses the incremental effect of more aggressive vs standard therapy. In an attempt to expedite evidence development, many have turned to surrogate markers, intermediate end points in the biological pathway, as more readily measurable alternatives to clinical outcomes. However, surrogate end points often fail if they are not in the direct causal pathway of the effect of the intervention or if the intervention causes harm independent of the disease process.

Even though surrogate end points such as IMT and left ventricular hypertrophy have evidence associating them with both cardiovascular risk factors and outcomes,^{9,10} there is a paucity of evidence showing that changes in these markers will accurately predict future cardiovascular events.11 For example, the Antioxidant Supplementation in Atherosclerosis Prevention (ASAP) study found that vitamin supplementation delayed carotid IMT progression, but had no effect on reducing clinical events.12 Similarly, while cholesterol lowering has been well associated with carotid IMT regression, the ENHANCE trial recently reported that aggressive lipid lowering may not necessarily correlate with IMT reduction, at least in a small short-term evaluation.13 In addition, tighter blood glucose control has been associated with delayed IMT progression, yet the ACCORD trial found that more aggressive glycemic control led to an excess of deaths relative to standard management.14

Without doubt, these trials underscore the need to look not just at surrogate markers, but at the ultimate clinical expression of the intervention—patient outcomes. However, there are no easy solutions for improving the throughput of trials with hard clinical end points. Enriching the study population with high-risk populations can increase event rates, but this is not always predictable (as evidenced by the unexpectedly low event rates in SANDS), and doing so can limit study generalizability. Longer and larger studies are ideal, yet using current trial methodologies, can be quite costly. SANDS,⁷ with only 500 patients, cost approximately \$12 million, whereas a 10 000-patient clinical end point–driven trial may cost in excess of \$200 million. Thus, methods to improve trial efficiency with better tools to identify eligible patients, quicker site activation, streamlined electronic data collection, and remote monitoring are needed to make trials more practical and affordable.

What are the take-home messages from SANDS? For the true believers, the study confirms that aggressive lipid and hypertension treatment has a favorable effect on proven "early markers" of disease. Thus, with longer duration of follow-up (which will hopefully be the case), the study would most assuredly demonstrate improved patient outcomes. For the therapeutic nihilists, however, SANDS took high-risk patients with type 2 diabetes, studied them under idealized circumstances, and still found no clinical benefit after 3 years of follow-up. In fact, an aggressive approach involved greater polypharmacy and costs and had a higher risk of adverse effects.

In contrast to these extremes and while awaiting longerterm data, a practical middle-of-the-road approach might be to support intensive lipid lowering with statin therapy in patients with diabetes, because this is supported by prior large, randomized, clinical, end point-driven trials,¹⁵ and has relatively few adverse effects or patient risks. For intensive blood pressure management, however, more data are needed because the benefits are not assured and there are modest, but measurement-negative effects on patients' finances and well-being. The blood pressure lowering group of the ACCORD trial,¹⁶ comparing a goal of lower than 120 mm Hg vs lower than 140 mm Hg, as well as the recently announced National Heart, Lung, and Blood Institute's Systolic Blood Pressure Intervention Trial (SPRINT), comparing aggressive vs standard blood pressure management in a large end point-driven trial, should provide these much needed data.

Finally, believers or not, all clinicians can support the pressing need to better assist patients in effectively modifying their risk factors to whatever goal deemed appropriate. Even in the rigorous National Heart, Lung, and Blood Institute's protocol-driven study environment of SANDS, with concomitant patient education and dedicated study physician and nursing care, blood pressure and lipid targets were reached in fewer than half of all patients. Now in community practice, only a third of patients with hypertension and hyperlipidemia are meeting standard treatment goals.¹⁷ This problem would only be magnified if future evidence mandates more stringent goals. Thus, novel strategies to facilitate patient engagement in their disease management are needed.^{18,19}

©2008 American Medical Association. All rights reserved.

(Reprinted) JAMA, April 9, 2008—Vol 299, No. 14 1719

EDITORIALS

In conclusion, SANDS is an important step forward in discovering whether lower goals are truly better for primary prevention. While the study results can be interpreted to support both viewpoints on the ideal target of therapy, such debates are healthy and will ultimately drive physicians to search for more definitive evidence as well as to seek systemwide strategies to effectively reach therapeutic goals in community practice.

Financial Disclosures: Dr Peterson reported receiving research support/grants from Schering-Plough, BMS/Sanofi, Merck/Schering-Plough, and St Jude, Inc. Dr Wang reported no financial disclosures.

REFERENCES

1. Cannon CP, Steinberg BA, Murphy SA, Mega JL, Braunwald E. Meta-analysis of cardiovascular outcomes trials comparing intensive versus moderate statin therapy. *J Am Coll Cardiol*. 2006;48(3):438-445.

2. Executive Summary of The Third Report of The National Cholesterol Education Program (NCEP) Expert Panel on Detection. Evaluation, and Treatment of High Blood Cholesterol In Adults (Adult Treatment Panel III). *JAMA*. 2001;285(19): 2486-2497.

3. Chobanian AV, Bakris GL, Black HR, et al. Seventh Report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure. *Hypertension*. 2003;42(6):1206-1252.

 Tonelli M, Isles C, Curhan GC, et al. Effect of pravastatin on cardiovascular events in people with chronic kidney disease. *Circulation*. 2004;110(12):1557-1563.

5. Kjekshus J, Apetrei E, Barrios V, et al. Rosuvastatin in older patients with systolic heart failure. *N Engl J Med*. 2007;357(22):2248-2261.

6. Sever PS, Dahlof B, Poulter NR, et al. Prevention of coronary and stroke events with atorvastatin in hypertensive patients who have average or lower-than-average cholesterol concentrations, in the Anglo-Scandinavian Cardiac Outcomes Trial—Lipid Lowering Arm (ASCOT-LLA): a multicentre randomised controlled trial. *Lancet.* 2003;361(9364):1149-1158.

7. Howard BV, Roman MJ, Devereux RB, et al. Effect of lower targets for blood pressure and LDL cholesterol on atherosclerosis in diabetes: the SANDS randomized trial. *JAMA*. 2008;299(14):1678-1689.

8. Sever P, Dahlof B, Poulter N, et al. Potential synergy between lipid-lowering

and blood-pressure-lowering in the Anglo-Scandinavian Cardiac Outcomes Trial. *Eur Heart J.* 2006;27(24):2982-2988.

9. O'Leary DH, Polak JF, Kronmal RA, Manolio TA, Burke GL, Wolfson SK Jr. Carotidartery intima and media thickness as a risk factor for myocardial infarction and stroke in older adults: Cardiovascular Health Study Collaborative Research Group. *N Engl J Med.* 1999;340(1):14-22.

10. Wachtell K, Okin PM, Olsen MH, et al. Regression of electrocardiographic left ventricular hypertrophy during antihypertensive therapy and reduction in sudden cardiac death: the LIFE Study. *Circulation*. 2007;116(7):700-705.

11. Bots ML, Baldassarre D, Simon A, et al. Carotid intima-media thickness and coronary atherosclerosis: weak or strong relations? *Eur Heart J*. 2007;28(4):398-406.

12. Salonen RM, Nyyssonen K, Kaikkonen J, et al. Six-year effect of combined vitamin C and E supplementation on atherosclerotic progression: the Antioxidant Supplementation in Atherosclerosis Prevention (ASAP) Study. *Circulation*. 2003; 107(7):947-953.

13. Merck/Schering-Plough Pharmaceuticals. Merck/Schering-Plough Pharmaceuticals provides results of the ENHANCE Trial. http://www.sch-plough .com/schering_plough/news/release.jsp?releaseID=1095943. Accessed March 12, 2008.

14. US Department of Health and Human Services; National Institutes of Health; National Heart, Lung, and Blood Institute. Action to Control Cardiovascular Risk in Diabetes (ACCORD) blood sugar treatment strategy announcement: February 6, 2008. http://www.nhlbi.nih.gov/health/prof/heart/other/accord/index.htm. Accessed March 12, 2008.

15. Heart Protection Study Collaborative Group. MRC/BHF Heart Protection Study of cholesterol lowering with simvastatin in 20,536 high-risk individuals: a randomised placebo-controlled trial. *Lancet*. 2002;360(9326):7-22.

16. Cushman WC, Grimm RH Jr, Cutler JA, et al. Rationale and design for the blood pressure intervention of the Action to Control Cardiovascular Risk in Diabetes (ACCORD) Trial. *Am J Cardiol.* 2007;99(12A)(supplement 1):S44-S55.

17. Rosamond W, Flegal K, Friday G, et al. Heart disease and stroke statistics— 2007 update: a report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2007;115(5):e69-171.

18. American Heart Association; American Stroke Association. American Heart Association Blood Pressure Management Center. https://bpmc.heart.org/. Accessed March 12, 2008.

19. Bosworth HB, Olsen MK, McCant F, et al. Hypertension Intervention Nurse Telemedicine Study (HINTS): testing a multifactorial tailored behavioral/educational and a medication management intervention for blood pressure control. *Am Heart J.* 2007;153(6):918-924.